Author's response to reviews

Title: Using molecular similarity to highlight the challenges of routine immunoassay-based drug of abuse/toxicology screening in emergency medicine

Authors:

Matthew D Krasowski (mdk24@pitt.edu)
Anthony F Pizon (pizonaf@upmc.edu)
Mohamed G Siam (mgsiam@hotmail.com)
Spiros Giannoutsos (giannoutso@upmc.edu)
Manisha Iyer (iyermanisha@gmail.com)
Sean Ekins (ekinssean@yahoo.com)

Version: 3 Date: 25 April 2009

Author's response to reviews:

4/24/2009

To Editor, BMC Emergency Medicine

Re: MS: 1521732942255134

We submit a revision of our manuscript entitled “Using molecular similarity to highlight the challenges of routine immunoassay-based drug of abuse/toxicology screening in emergency medicine” to address queries in the proofing stage. The specific revisions are as follows:

(1) “References 25, 26, In press – Have these articles been published? If so, please provide the citation details.”

Both articles are not published yet and citation details are not yet available.

(2) “Table vertical elements – please note that we are unable to display vertical rules within the tables – please re-layout your table without these vertical elements.”

This change has been made to Tables 1, 2, and 3.

(3) Additional files – Please format your additional files section like the example below...

This change has been made to the Additional Files section.

(4) “References – all references need to be cited in the main body of the manuscript text in order to ensure that they link correctly in the final publication. We notice you have cited reference 76 in the figure legend. Please include the reference in the text with the first citation for Figure 4. You may need to renumber for subsequent references.”
Reference 76 is now cited with the first citation for Figure 4 in the main text of the manuscript. This reference is now reference 54 with the other subsequent references re-numbered.

Thank for your help.

Sincerely,

Matthew D. Krasowski, MD, PhD
Assistant Professor
Department of Pathology
University of Pittsburgh
Scaife Hall S-737
3550 Terrace Street
Pittsburgh, PA 15261, USA
Phone: 412-647-6517
Fax: 412-647-5934
E-mail: mdk24@pitt.edu